Author's response to reviews

Title: Factors associated with excessively lengthy treatment of tuberculosis in the eastern Paris region of France in 2004.

Authors:

Nadia Valin (nadia.valin@sat.aphp.fr)
Gilles Hejblum (gilles.hejblum@u707.jussieu.fr)
Isabelle Borget (isabelle.borget@igr.fr)
Henri-Pierre Mallet (henri-pierre.mallet@sante.gov.pf)
Fadi Antoun (Fadi.Antoun@paris.fr)
Didier Che (d.che@invs.sante.fr)
Christos Chouaid (christos.chouaid@sat.aphp.fr)

Version: 4 Date: 9 August 2010

Author's response to reviews: see over
Dear Editorial Team,

Thank you for your letter (9-August-2010) concerning our manuscript MS: 3365521693538275. According to your comments, we made the following changes in the final version of the manuscript:

Major revisions

1. We ensure that the abstract present in the manuscript matches exactly what is in the submission system.
2. We included the two tables in the manuscript in the tables section following the references.

Minor revisions

1. We removed all tracking from the manuscript
2. Introduction was replaced by Background
3. We did not modify the Figure

We hope that these changes conform to journal style.

Sincerely Yours,

Nadia Valin